Author's response to reviews

Title: Precursor T-cell acute lymphoblastic leukemia presenting with bone marrow necrosis: a case report

Authors:

   Najmaddin S.H. Khoshnaw (najmaddin_salih@yahoo.com)
   Hisham A. Al-Rawi (haalrawi@yahoo.com)
   Beston F. Nore (beston.nore@univsul.net)

Version: 3 Date: 16 August 2012

Author's response to reviews: see over
Dear Editor,

Please find enclosed our revised manuscript entitled “Precursor T-Cell Acute Lymphoblastic Leukemia presenting with bone marrow necrosis: We are now introduced all changes requested by the editor and the reviewer, as following:

1. As requested by the reviewer, we introduced a new paragraph, end of the introduction section, describing BMN prognosis in four separate reports on the pediatric ALL.
2. We replaced boy by male and delete the hospital name.
3. We delete the web-link in the consent section.
4. The new sentence is added in the authorship contributions section.
5. We did some additions/corrections in the acknowledgements.
6. Reference list is updated according to the EndNote-biomedcentral formatting.
7. As requested, Figure 3 is revised to protect the patient identity.
8. We corrected some minor errors in the text and in the figure legends.
9. We adjust comment about prognosis, as requested in the latest comments.

We hope now that all changes according to the requests and the manuscript will be accepted for publication. We are looking forward to hear from you.

Kind regards /

Dr. Najmaddin Khoshnaw

MD, HDH, Head of the Haematology Department,
Hiwa Hematology-Oncology Hospital,
Orga /318/Al-Sulaimaniyah province
Kurdistan Regional Government, IRAQ
Mobile: 00964 770 155 4184
eMail: najmaddin_salih@yahoo.com